

Author's response to reviews

Title: Discriminating lymphomas and reactive lymphadenopathy in lymph node biopsies by gene expression profiling

Authors:

To Ha Loi (t.loi@amr.org.au)
Anna Campain (annac@maths.usyd.edu.au)
Adam Bryant (a.bryant@amr.org.au)
Tim J Molloy (t.molloy@amr.org.au)
Mark Lutherborrow (m.lutherborrow@amr.org.au)
Jennifer Turner (jtur8838@bigpond.net.au)
Yee Hwa Jean Yang (jeany@maths.usyd.edu.au)
David DF Ma (d.ma@amr.org.au)

Version: 2 **Date:** 17 August 2010

Author's response to reviews: see over

Blood, Stem Cell & Cancer Research Unit
Department of Haematology,
St Vincent's Hospital
390 Victoria St
Darlinghurst NSW 2010 Australia

17th August, 2010

Dear Editor of BMC Medical Genomics,

Re: Submission of revised version of manuscript MS: 1852645016427570
“Discriminating lymphomas and reactive lymphadenopathy in lymph node biopsies by
gene expression profiling”

As requested, the microarray data of this study have been deposited into GEO, a public repository for microarray data, with series number GSE23647. Our manuscript has been altered accordingly with this new information.

The following link has been created to allow private review of record GSE23647:
<http://www.ncbi.nlm.nih.gov/geo/query/acc.cgi?token=rvkfjeigsoscini&acc=GSE23647>

Please let me know if there are any problems accessing the link.

Thank you and kind regards,

David D.F. Ma
Professor of Haematology